Author's response to reviews

Title: Behavioural symptoms in patients with Alzheimer's disease and their association with cognitive impairment.

Authors:

COOPERA Study Group (COOPERA) (mfernandezm@meditex.es)
Manuel Fernández (mfernandezm@meditex.es)

Version: 5 Date: 7 June 2010

Author's response to reviews: see over
Dear editor and reviewer:

We strongly appreciate the effort you have undertaken in reviewing the article entitled “Behavioural symptoms in patients with Alzheimer’s disease and their association with cognitive impairment”.

Please find enclosed the manuscript which we have modified according to the major and minor revisions that you proposed. First, we have included author’s names in the acknowledgments, so that only the name of the study group is in the title page section. Second, the group contribution has been given in the author’s contribution section. As for minor revisions, we removed author’s qualifications and job titles from the manuscript, we amended capitalization of the author list, and we modified abbreviation section by formatting into a paragraph. Besides, we move the statement: “The article is not under consideration elsewhere and it has been read and approved by all listed authors”, which was in the competing interest section, to the authors’ contribution section.

Sincerely yours

M. Fernandez on behalf of all authors.